Author's response to reviews

Title: Development of a Blood-based Gene Expression Algorithm for Assessment of Obstructive Coronary Artery Disease in Non-Diabetic Patients

Authors:

Michael R Elashoff (melashoff@cardiodx.com)
James A Wingrove (jwingrove@cardiodx.com)
Philip Beineke (pbeineke@cardiodx.com)
Susan E Daniels (sdaniels@cardiodx.com)
Whittemore G Tingley (tingley.whittemore@gene.com)
Steven Rosenberg (srosenberg@cardiodx.com)
Szilard Voros (szilard.voros@piedmont.org)
William E Kraus (william.kraus@duke.edu)
Geoffrey S Ginsburg (geoffrey.ginsburg@duke.edu)
Robert S Schwartz (robert.stockton.schwartz@gmail.com)
Stephen G Ellis (elliss@ccf.org)
Naheem Tahirkheli (ntahirkheli@ocaheart.com)
Ron Waksman (ron.waksman@medstar.net)
John McPherson (john.mcpherson@vanderbilt.edu)
Alexandra J Lansky (alexandralansky@gmail.com)
Eric J Topol (etopol@scripps.edu)

Version: 4 Date: 22 March 2011

Author's response to reviews:

BMC Editorial Production Group,

I have made all of the major and minor revisions suggested. In particular, what was Table 1A has become Table 1 and Table 1B has become Tables 3, as it is referred to after Table 2. Please check that the new Table formats are acceptable. Thank you for your help.

Steve Rosenberg, Corresponding author